Author's response to reviews

Title: Nuclear location of tumor suppressor protein maspin inhibits proliferation of breast cancer cells without affecting proliferation of normal epithelial cells

Authors:

Magdalena Machowska (mmachowska@ibmb.uni.wroc.pl)
Katarzyna Wachowicz (Katarzyna.Wachowicz@i-med.ac.at)
Miroslaw Sopel (mirek.sopel@gmail.com)
Ryszard Rzepecki (rzepecki@ibmb.uni.wroc.pl)

Version: 5 Date: 29 January 2014

Author's response to reviews:

Dear Editor,

According to Your suggestion we added the info that our study was approved by Bioethical Committee.

We added the following sentence into the Materials and Methods section:
Bioethical Committee of the Lower Silesian Oncology Center approved the study.

Kind regards
Ryszard Rzepecki